Author's response to reviews

Title: Effect of an Electronic Nicotine Delivery Device (e-Cigarette) on Smoking Cessation and Reduction: A Prospective Pilot Study

Authors:

Riccardo Polosa (polosa@unict.it)
Pasquale Caponnetto (p.caponnetto@unict.it)
Jaymin B Morjaria (J.B.Morjaria@soton.ac.uk)
Gabriella Papale (gabriellapapale@virgilio.it)
Davide Campagna (davidecampagna83@gmail.com)
Cristina Russo (kristina_russo@yahoo.com)

Version: 7 Date: 22 August 2011

Author's response to reviews: see over
Dear Sir,

Re.: Formatting changes due for MS 1187998564873930 - Effect of an Electronic Nicotine Delivery Device (e-Cigarette) on Smoking Reduction and Cessation: A Prospective 6-Month Pilot Study

As per your request, we have now included the following amendments:
Title and Abstract have been made identical to the ms in the submission system;
Abstract replaces Summary;
A Conclusion paragraph has been added;
Authors’ contributions section has been amended in to a single paragraph;
Authors’ titles have been removed from the authors list in the Title page;
Link to website has been checked (it is working fine now);
Tables (three tables) have been included in the manuscript – accordingly the three additional files uploaded as table 1 to 3 have been removed from the system;
Figures 3A, 3B and 3C have been renamed as Figures 3, 4 and 5 (text citation and legends have been corrected accordingly);
Additional files 4 to 7 have been removed

Yours sincerely, Riccardo Polosa